Author’s response to reviews

Title: Acceptance of interprofessional learning between medical and pharmacy students in a prescribing skills training workshop: Pre-post intervention study

Authors:

Siew Siang Chua (SiewSiang.Chua@taylors.edu.my)

Pauline Siew Mei Lai (plai@ummc.edu.my)

Si Mui Sim (debrasim@ummc.edu.my)

Choo Hock Tan (tanch@um.edu.my)

Chan Choong Foong (foongchanchoong@um.edu.my)

Version: 1 Date: 10 Nov 2017

Author’s response to reviews:

1) Information regarding the study was provided to all the participants and participation in the questionnaire study was voluntary. Written informed consent from the participant was not obtained as the prescribing skills workshop was part of the undergraduate medical and pharmacy curricula. This exemption of obtaining written informed consent was approved by the Medical Research Ethics Committee, University of Malaya Medical Centre (MREC ID NO: 20168-2647). The ethics and consent statement in the manuscript has been amended accordingly.

2) Corresponding author's email address has been amended to be identical as with the online submission.

3) Authors' names are given in full initials in the Authors' Contributions section.

4) Consent to Publish is included.